Author’s response to reviews

Title: Associations of BAFF rs2893321 polymorphisms with myasthenia gravis susceptibility

Authors:
Hui Deng (huideng_ny@163.com)
Jianjian Wang (wangjj_hrb@163.com)
Xiaotong Kong (xiaotongk_23@163.com)
Huixue Zhang (huixue81@163.com)
Tianfeng Wang (wg33tianfeng@163.com)
Wenqi Tian (wqtian665@163.com)
Tingting Yi (yting2g@163.com)
Lihua Wang (lihua_wlh@163.com)

Version: 3 Date: 25 Sep 2019

Author’s response to reviews:

1. Availability of data and materials

Your current statement in the Availability of data and materials section is incorrect. Please provide the relevant accession numbers if the SNP genotyping data has been deposited into a data bank or the appropriate web links if the data has been uploaded into a repository. If the data has not yet been deposited, please do so and provide the relevant information needed to access it in the ‘Availability of data and materials’ section in your manuscript. Please ensure that all data is already released and that any accession numbers/web links are in the ‘Availability of Data and Materials’ in their final form, and that all web links and accession numbers allow access to public data.

ANSWER: The SNP genotyping in this study was detected in our sample using iMLDR technology. We haven’t upload the data in any online data bank. We have changed the statement into “Availability of data and materials section” into “The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.”

2. Declarations heading

Please ensure that you include a ‘Declarations’ heading and that it is placed after the Abbreviations in
your manuscript.
The Declarations section encompasses the sub-sections: Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors’ contributions, Acknowledgements.

ANSWER: we have revised the heading.

3. Authors’ contributions

Please amend the initials for Hui Deng from “DH” to “HD”.
Please amend the initials for Jianjian Wang from “WJJ” to “JJW”.
Please amend the initials for Xiaotong Kong from “KXT” to “XTK”.
Please amend the initials for Huixue Zhang from “ZHX” to “HXZ”.
Please amend the initials for Tianfeng Wang from “WTF” to “TFW”.
Please amend the initials for Wenqi Tian from “TWQ” to “WQT”.
Please amend the initials for Tingting Yi from “YTT” to “TTY”.
Please amend the initials for Lihua Wang from “WLH” to “LHW”.

ANSWER: we have revised the initials.

4. Consent for publication

Currently, the statement in your “Consent for publication” section of your declarations is incorrect. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

ANSWER: we have revised the Consent for publication section.